Author's response to reviews

Title: The influence of a series of five dry cupping treatments on pain and mechanical thresholds in patients with chronic non-specific neck pain - a randomised controlled pilot study

Authors:

Romy Lauche (r.lauche@kliniken-essen-mitte.de)
Holger Cramer (h.cramer@kliniken-essen-mitte.de)
Kyung-Eun Choi (k.choi@kliniken-essen-mitte.de)
Thomas Rampp (t.rampp@kliniken-essen-mitte.de)
Felix Joyonto Saha (f.saha@kliniken-essen-mitte.de)
Gustav J Dobos (g.dobos@kliniken-essen-mitte.de)
Frauke Musial (frau.ek.musial@uit.no)

Version: 4 Date: 13 August 2011

Author's response to reviews: see over
Dear Editor, dear Reviewers

We are very delighted that you accepted the manuscript. Please find attached the revised version of the manuscript "The influence of a series of five dry cupping treatments on pain and mechanical thresholds in patients with chronic non-specific neck pain - a randomised controlled pilot study".

The following format changes have been made:
1. Affiliations: they were changed according to the expected format.
2. Trial registration: Was added to the abstract.
3. Contribution: The final manuscript was checked by all co-authors. It was added in the contributions.
4. All highlighting and tracking were removed.
5. Reference with more than 30 authors was changed using et al.
6. Tables with merged cells across rows were re-layouted.
7. Figures were submitted a separate files. We used the original powerpoint graphics to achieve highest resolution.
8. Manuscript has again been checked for grammar and spelling.

Sincerely yours

Romy Lauche